Author's response to reviews

Title: Perforating eyelid injury extending to the brain stem in a 17-year-old woman: a case report

Authors:

Eiichiro Noda (noda@eye-center.org)
Makoto Inoue (inoue@eye-center.org)
Izumi Kobayashi (kobayashi@eye-center.org)
Toshiyuki Nagamoto (nagamoto@eye-center.org)

Version: 2 Date: 13 January 2010

Author's response to reviews: see over
Dear Sir,

Thank you for the acceptance of our manuscript. The manuscript has been improved based on your comments. Changes to the manuscript are indicated below.

Reference 6 has been cited in the manuscript text but not included in the reference list. Please amend this.

Answer: We added reference 6 in the list.

Please move the consent section to a position directly before the competing interests section.

Answer: We moved the consent section to a position directly before the competing interests section.